Author's response to reviews

Title: Academic Leagues: a Brazilian way to teach about cancer in medical universities.

Authors:

    Diogo AV Ferreira (dii.valente@gmail.com)
    Renata N Aranha (renataaranhauerj@yahoo.com.br)
    Maria HF Ornellas (mariahelenaornellas@gmail.com)

Version: 2  Date: 4 May 2015

Author's response to reviews:

Dear editors,

In this review, the manuscript has been updated with the line numbers and ethical considerations.

Best wishes

Diogo Antonio Valente Ferreira